Author's response to reviews

Title: Correlates of cognitive susceptibility to smoking among Mexican origin youth residing in Houston, Texas

Authors:

Anna V Wilkinson (awilkins@mdanderson.org)
Andrew J Waters (andrew.waters@usuhs.mil)
Vandita Vasudevan (vvasudev@mdanderson.org)
Melissa L Bondy (mbondy@mdanderson.org)
Alexander V Prokhorov (aprokhor@mdanderson.org)
Margaret R Spitz (mspitz@mdanderson.org)

Version: 6 Date: 17 September 2008

Author's response to reviews: see over
September 17th 2008

Dear Editors:

I made three minor changes to the text, but do not see other typographical errors. I hope I caught them all.

Thank you,

Anna Wilkinson